Author's response to reviews

Title: Supporting adolescent emotional health in schools: a mixed methods study of student and staff views in England

Authors:

Judi Kidger (judi.kidger@bristol.ac.uk)
Jenny Donovan (jenny.donovan@bristol.ac.uk)
Lucy Biddle (lucy.biddle@bristol.ac.uk)
Rona Campbell (rona.campbell@bristol.ac.uk)
David Gunnell (D.J.Gunnell@bristol.ac.uk)

Version: 4 Date: 28 October 2009

Author's response to reviews:

Revisions made to text

1. Footnotes have been removed and replaced with text
2. Boxes have been changed to tables, and references to them in the text changed accordingly
3. Title has been changed to sentence case
4. Text been checked for typos, a few words changed.
5. Formatting of tables has been tidied up